Author's response to reviews

Title: Genomic imbalances in 5918 malignant epithelial tumors: An explorative meta-analysis of chromosomal CGH data.

Authors: 
Michael Baudis (mbaudis@gmail.com)

Version: 2 Date: 26 April 2007

Author's response to reviews:

The figure numbering was changed. The original version had 1A/1B, which did not correspond to the automatic numbering of the uploaded files. No other changes.